Author's response to reviews

Title: Radiation-induced morphea of the breast: a case report

Authors:

Nellie LC Cheah (nelliecheah@yahoo.co.uk)
Daniel WY Wong (danielwywong@hotmail.com)
Anula D Chetiyawardana (Shan.Chetiyawardana@uhb.nhs.uk)

Version: 5 Date: 1 April 2008

Author's response to reviews:

Thank you for your e-mail below.

There are 3 amendments that I would like to make:

1. In page 3, paragraph 3, line 5, the sentence
   '...infiltrate consisting of lymphocytes, predominantly plasma cells and histiocytes (Figure 2).'
   is amended to
   'infiltrate consisting of lymphocytes predominantly, plasma cells and histiocytes (Figure 2).'
   ie the comma sign should be placed after the word 'predominantly' and not after the word 'lymphocytes'.

2. In page 4, paragraph 1, line 4, the sentence
   '...3.5 years post-radiotherapy (Figures 1, right panel). ...'
   is amended to
   '...3.5 years post-radiotherapy (Figure 1, right panel). ...'
   ie there is no letter 's' in the word 'Figure'.

3. In page 4, paragraph 2, line 1, the sentence
   'RIM of the breast is rare, with an estimated incidence of 0.20% (see [4]).'
   is amended to
   'RIM of the breast is rare, with an estimated incidence of 0.20% [4].'
   ie (see...) is deleted.

With the above 3 amendments added, I confirm that I agree with all the amendments made in the manuscript. The manuscript with the 3 amendments added is enclosed.

I look forward to hearing from you. Thank you.
Yours sincerely,
Dr Nellie Cheah